\documentclass[CRMATH,Unicode,biblatex,XML,published]{cedram}

\TopicFR{Équations aux dérivées partielles}
\TopicEN{Partial differential equations}

\addbibresource{CRMATH_Baghaei_20230713.bib}


\usepackage{amssymb,amscd}
\usepackage{mathtools}

\numberwithin{equation}{section}


\newcommand{\dd}{\mathrm{d}}
\DeclareMathOperator{\expe}{e}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}


\newcommand*{\romanenumi}{\renewcommand*{\theenumi}{\roman{enumi}}}
\newcommand*{\Romanenumi}{\renewcommand*{\theenumi}{\Roman{enumi}}}
\newcommand*{\alphenumi}{\renewcommand*{\theenumi}{\alph{enumi}}}
\newcommand*{\Alphenumi}{\renewcommand*{\theenumi}{\Alph{enumi}}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\title{Global boundedness of solutions to a chemotaxis consumption model with signal dependent motility and logistic source}
\alttitle{Limite globale des solutions d'un modèle de consommation de chimiotaxie avec motilité dépendante du signal et source logistique}

\author{\firstname{Khadijeh} \lastname{Baghaei}}

% \address{\textcolor{red}{Address is missing}}

\thanks{This research was supported by a grant from 
PIAIS (No. 1402-10108).}

\CDRGrant[PIAIS]{1402-10108}


%\keywords{\textcolor{red}{missing}}
%\altkeywords{\textcolor{red}{missing}}
%\subjclass{\textcolor{red}{missing}}



 \begin{abstract}
 This paper deals with the following chemotaxis system:
 \begin{equation*}
	\begin{cases}
	u_{t}=\nabla\cdot\big(\gamma(v) \,\nabla u-u \,\xi(v) \,\nabla v\big)+\mu\, u\,(1-u), & x\in\Omega, \ t>0, \\
	v_{t}=\Delta v-uv, & x\in\Omega, \ t>0,
	\end{cases}
	\end{equation*}
 under homogeneous Neumann boundary conditions
 in a bounded domain 
 $ \Omega \subset \mathbb{R}^{n}, n\geq 2,$ 
 with smooth boundary.
Here, the functions $\gamma(v)$ and
$\xi(v)$ are as:
\begin{equation*}
 \gamma(v)=(1+v)^{-k}\quad
 \mbox{and} \quad
 \xi(v)=-(1-\alpha)\,\gamma'(v),
\end{equation*}
where $k>0$ and $\alpha \in (0,1).$\\
 For the above system, we prove that 
 the corresponding initial boundary value problem admits a unique global
 classical solution which is
 uniformly-in-time bounded.
This result is obtained under some conditions on initial value $ v_{0}$ 
 and $\mu$ and without any restriction on $k$ and $\alpha.$
 The obtained result extends the recent results obtained for this problem.
 
\end{abstract}


\begin{altabstract}
 Cet article porte sur le système de chimiotaxie suivant :
 \begin{equation*}
	\begin{cases}
	u_{t}=\nabla\cdot\big(\gamma(v) \,\nabla u-u \,\xi(v) \,\nabla v\big)+\mu\, u\,(1-u), & x\in\Omega, \ t>0, \\
	v_{t}=\Delta v-uv, & x\in\Omega, \ t>0,
	\end{cases}
	\end{equation*}
 sous des conditions aux limites homogènes de Neumann dans un domaine borné
 $ \Omega \subset \mathbb{R}^{n}, n\geq 2,$ 
 avec une frontière lisse. Ici, les fonctions $\gamma(v)$ et
$\xi(v)$ sont les suivantes :
\begin{equation*}
 \gamma(v)=(1+v)^{-k}\quad
 \mbox{and} \quad
 \xi(v)=-(1-\alpha)\,\gamma'(v),
\end{equation*}
où $k>0$ and $\alpha \in (0,1).$\\
 Pour le système ci-dessus, nous prouvons que le problème de valeur limite initiale correspondant admet une unique solution classique globale qui est uniformément bornée en temps.
Ce résultat est obtenu sous certaines conditions sur la valeur initiale $ v_{0}$ et $\mu$ et sans restriction sur $k$ et $\alpha.$
Le résultat obtenu étend les résultats récents obtenus pour ce problème. 
\end{altabstract}


\dateposted{2024-11-05}
\begin{document}
\begin{thebibliography}{99}
\bibitem{ks}
{
E. F. Keller, L. A. Segel, Initiation of slime mold aggregation viewed as an instability, J. Theor. Biol.,} {26} (1970) 399–415.

\bibitem{24}{
Y. Xiao, J. Jiang, Global existence and uniform boundedness in a fully
parabolic Keller–Segel system with non-monotonic
signal-dependent motility, J. Differ. Equ., }{354} 
(2023) 403-429.


\bibitem{Yk}
 {C. Yoon, Y.-J. Kim, Global existence and aggregation in a Keller--Segel model with Fokker-Planck diffusion, Acta Appl. Math.,} {149} (2017) 101-123.

\bibitem{AY}
 { J. Ahn, C. Yoon, Global well-posedness and stability of constant equilibria in parabolic–elliptic chemotaxis systems without gradient sensing, Nonlinearity,} {32} (2019) 1327.
 
\bibitem{FS}
 {
K. Fujie, T. Senba, Global boundedness of solutions to a parabolic–parabolic chemotaxis system with local sensing in higher dimensions, Nonlinearity,} {35} (2022) 3777.

\bibitem{2020}
 {
H.-Y. Jin, Z.-A. Wang, Critical mass on the Keller-Segel system with signal-dependent motility, Proc. Am. Math. Soc.,} {148} (2020) 4855-4873.

\bibitem{Fj}
 {
K. Fujie, J. Jiang, Global existence for a kinetic model of pattern formation with density-suppressed motilities, J. Differ. Equ.,} {269} (2020) 5338–5378.

\bibitem{2021}
 {
K. Fujie, J. Jiang, Boundedness of classical solutions to a degenerate Keller-Segel type model with signal-dependent motilities, Acta Appl. Math.,} {176} (2021) 3.

\bibitem{20199}
 {L. Desvillettes, Y. J. Kim, A. Trescases, and C. W. Yoon, A logarithmic chemotaxis model featuring global existence and
aggregation, Nonlinear Anal. Real World Appl.,} {50} (2019), 562–582.


\bibitem{2021_1}
 {M. Burger, L. Philippe and T. Ariane, Delayed blow-up for chemotaxis models with local
sensing, J. London Math. Soc.,} (2){103} (2021) 1596–1617. 

\bibitem{55} {
H.-Y. Jin, Y.-J. Kim, Z.-A. Wang, Boundedness, stabilization, and pattern formation driven by density-suppressed motility, SIAM J. Appl. Math.,} {78} (2018) 1632–1657.

\bibitem{2019}
 {
 J. Wang, M. Wang, Boundedness in the higher-dimensional Keller-Segel model with signal-dependent motility and logistic growth, J. Math. Phys.,} {60} (2019) 011507.

 \bibitem{winkler2022}
{ M. Winkler, A Result on Parabolic Gradient Regularity in Orlicz Spaces
and Application to Absorption-Induced Blow-Up Prevention in
a Keller–Segel-Type Cross-Diffusion System,
International Mathematics Research Notices,} Vol. 2023, No. 19, 16336–16393.


 \bibitem{2005}
{I. Tuval, L. Cisnerous, C. Dombrowski, C. W. Wolgemuth, J. O. Kessler, R. E. Goldstein, Bacterial swimming and oxygen transport near contact lines, Proc.
Natl. Acad. Sci.,} USA 102 (2005) 2277–2282.


\bibitem{2021w}
{T. Black, M. Winkler, Global weak solutions and absorbing sets in a chemotaxis-Navier-Stokes system
with prescribed signal concentration on the boundary, Math. Models Methods Appl. Sci.,} {32} (2022) 137-173.

\bibitem{2014w}
{M. Winkler,
Stabilization in a two-dimensional chemotaxis-Navier-Stokes system, Arch. Ration. Mech.
Anal.,} {211}(2) (2014)
 455–487.
 
\bibitem{2016l}
{X. Cao, J. Lankeit, Global classical small-data solutions for a three-dimensional chemotaxis Navier-Stokes system involving matrix-valued sensitivities, Calc. Var.} {55} (2016) 55:107.

\bibitem{2023}
{J. Han, C. Liu, Global weak solution for a chemotaxis Navier-Stokes system with
p-Laplacian diffusion and singular sensitivity,
Nonlinear Analysis: Real World Applications,}
{73} (2023) 103898.

\bibitem{taowin}
{Y. Tao, M. Winkler, Eventual smoothness and stabilization of large-data solutions in a three-dimensional chemotaxis system with consumption of chemoattractant, J. Differ. Equ.,} {252} (2012) 2520–2543.

\bibitem{tao}
{Y. Tao, Boundedness in a chemotaxis model with oxygen consumption by bacteria, J. Math. Anal. Appl.,} {381} (2011) 521–529.

\bibitem{baghaei-khelghati}
{K. Baghaei, A. Khelghati, Boundedness of classical solutions for a chemotaxis model with consumption of chemoattractant, C. R. Acad. Sci. Paris, Ser. I,}
{355} (2017) 633–639.

\bibitem{baghaei-khelghati-logistic}
{K. Baghaei, A. Khelghati, 
Global existence and boundedness of classical solutions for a chemotaxis model with consumption of chemoattractant and logistic source, Mathematical Methods in the Applied Sciences,} {40} (2017)
 3799-3807.

 \bibitem{GW}
{G. Li, M. Winkler, Relaxation in a Keller-Segel-consumption system
involving signal-dependent motilities,
Communications in Mathematical Sciences,} {21} (2) 299–322.

\bibitem{ph}
{P. Lauren\c{c}ot, Long term spatial homogeneity for a chemotaxis model with local sensing and
consumption, Communications in Mathematical Sciences,} {21} (6) (2023) 1743–1750. 


\bibitem{6}
{
D. Li, J. Zhao, Global boundedness and large time behavior of solutions to a chemotaxis-consumption system with signal-dependent motility, Z. Angew. Math. Phys.,} {72} (2021) 57.

\bibitem{app}
{G. Li, M. Winkler, Refined regularity analysis for a Keller-Segel-consumption system involving signal-dependent motilities, Appl. Anal., }


\bibitem{w}
{ M. Winkler, Global generalized solvability in a strongly degenerate taxis-type parabolic system
modeling migration–consumption interaction, Z. Angew. Math. Phys.,} (2023) 74:32.

\bibitem{tw2023}
{Y. Tao, M. Winkler, Global solutions to a Keller-Segel-consumption system
involving singularly signal-dependent motilities
in domains of arbitrary dimension,
J. Diff. Equ.,} {348} (2023) 390–418.


\bibitem{5}
{X. Li, L. Wang, X. Pan, Boundedness and stabilization in the chemotaxis consumption model with signal dependent motility,
Z. Angew. Math. Phys.,} {72} (2021) 170.


\bibitem{7}
{L. Wang,
Global dynamics for a chemotaxis consumption system with signal-dependent motility and logistic source,
J. Diff. Equ.,} {348} (2023) 191–222.



\bibitem{4}
{Y. Li, S. Lu, Global solutions to a chemotaxis-growth system with signal-dependent motilities and signal consumption,
J. Math. Anal. Appl.,} {521} (2023) 126902.

\bibitem{KHB}
{K. Baghaei, Boundedness of classical solutions to a chemotaxis consumption
system with signal dependent motility and logistic source, C. R. Acad. Sci. Paris, Ser. I,}
{361} (2023) 1641-1652.

 \bibitem{KB1}
 {A. Khelghati, K. Baghaei,
Boundedness of classical solutions for a chemotaxis system with general sensitivity function, 
Applicable Analysis,} {98} (2019) 611-621.

 \bibitem{KB2}
 {A. Khelghati, K. Baghaei,
Boundedness of classical solutions for a chemotaxis model with rotational flux terms, ZAMM} {98} (2018) 1864-1877.




\end{thebibliography}

\maketitle


\section{Introduction}
In this paper, we consider the following initial boundary value problem:
\begin{equation}\label{yazahraadrecni}
	\begin{cases}
	u_{t}=\nabla\cdot\big(\gamma(v)\, \nabla u-u \,\xi(v)\, \nabla v\big)+\mu \, u\,(1-u), & x\in\Omega, \ t>0, \\
	v_{t}=\Delta v-uv, & x\in\Omega, \ t>0,\\
	 \frac{\partial u}{\partial \nu}= \frac{\partial v}{\partial \nu}= 0,& x\in \partial\Omega, \ t>0, \\
	u (x , 0)= u_{0}, \ v(x ,0)= v_{0}, & x\in\Omega,
	\end{cases}
	\end{equation}
where $ \Omega \subset \mathbb{R}^{n}, n\geq 2,$ is a bounded domain with smooth boundary, $\nu$ denotes the unit outward normal vector to $\partial \Omega$ and $u_{0}$ and $v_{0}$ are initial functions. In the above problem,
$u=u(x, t)$ is the cell density and $v=v(x, t)$ denotes the nutrient consumed chemical concentrations.\\
In mathematical biology, systems such as~\eqref{yazahraadrecni} describe the mechanism of chemotaxis. The chemotaxis is the movement of cells towards a higher concentration
a chemical signal substance produced by cells.
We first state the results related to the classical chemotaxis system, which has been introduced by Keller and Segel in~\cite{ks}. The classical chemotaxis system can be written as follows:
\begin{equation}\label{145}
	\begin{cases}
	u_{t}=\nabla\cdot\big(\gamma(v) \,\nabla u-u \,\xi(v)\, \nabla v\big)+\mu\, u\,(1-u), & x\in\Omega, \ t>0, \\
	\tau v_{t}= \Delta v - v + u, & x\in\Omega, \ t>0, 
	\end{cases}
	\end{equation}
 where $\tau \in \{0,1\}.$
 In the following, we write some important results for this problem in the absence and presence of the logistic source, respectively. In the absence of logistic source,
when $\gamma \in C^{3}((0, \infty))$ and 
$ \xi=-\gamma'$ as well as
\begin{align*}
 \limsup _{s \rightarrow \infty} \gamma(s) < \frac{1}{\tau},
\end{align*}
then the problem~\eqref{145} admits a unique global classical solution in any dimension~\cite{24}. Also, for $n \geq 1,$
when the function $\gamma$ has strictly positive
 upper and lower bounds, then the classical solutions are 
 uniformly-in-time bounded~\cite{24}. 
 This result is also proved
 when $n \geq 2$ and the function $\gamma$
 decays at a certain slow rate
at infinity~\cite{24}. 
Among the special functions that have been studied as $\gamma$ are: 
\begin{align*}
 & \gamma(v)= c_{0} \, v^{-k}\quad \mbox{with} \quad k>0 \quad \mbox{and} \quad c_{0} >0
 \end{align*}
 and
\begin{align*} 
 & \gamma(v)=\expe^{-\chi v}\quad \mbox{with} \quad \chi>0.
\end{align*}
For $\gamma(v)= c_{0} \, v^{-k},$ 
 for all $k>0,$ 
 the global existence and boundedness of the solution is proved
 under a smallness assumption on $c_{0}$ in any dimension~\cite{Yk}.
 Also, for this function, by removing the assumption on $c_{0},$ 
 when $n \geq 2$ and $k \in (0, \frac{2}{n-2}),$ the same result is proved in cases $\tau=0$ 
~\cite{AY} and $\tau=1$~\cite{FS}.
 In the other special case $\gamma(v)= \expe^{-\chi v},$ if $n=2$ and
 $\int_{\Omega} u_{0} \, \dd x < \frac{4 \pi}{\chi},$ then the classical solutions are global and bounded whereas for $\int_{\Omega} u_{0} \, \dd x > \frac{4 \pi}{\chi},$
 blow up occurs either in finite or infinite time~\cite{2020}. But, in the case of
 $\tau=0,$
 the blow up occurs in infinite time~\cite{Fj}.
 Also, for $n=2,$ the classical
solutions are globally bounded if the positive function $\gamma$ decreases slower than an exponential
speed at high signal concentrations~\cite{2021}. For $n \geq 3,$ the same result is true when
$\gamma$ decreases at certain algebraically speed~\cite{2021}.
We now state the result obtained in
\cite{20199} which is the global existence of very weak solution to the problem~\eqref{145} when $\gamma(v)=\frac{1}{c+v^{k}}$ with $c\geq 0$ and $k>0.$ This result was obtained without any smallness assumption on the initial data provided that
\begin{equation*}
	 k\in 
	 	\begin{cases}
	 (0, \frac{7}{3}), &\mbox{if } n=1, \\
	(0,2), &\mbox{if } n=2,\\
(0,\frac{4}{3}), & \mbox{if } n=3.
	\end{cases}
	\end{equation*}
 Also, for $n\geq 1$ with $\gamma(v)=\expe^{-v},$ the existence of weak-strong solutions is proved 
 in
~\cite{2021_1}.
We now state the results to the problem~\eqref{145} in the presence of logistic source.
 If the decreasing function $\gamma \in C^{3}([0, \infty))$ satisfies
$\lim _{s \rightarrow \infty}\gamma(s)=0 $
and $\lim _{s \rightarrow \infty}\frac{\gamma'(s)}{\gamma(s)} $ exists, then the classical solutions are global and bounded for $n=2$~\cite{55}.
For $n \geq 3,$ the same result is true when $\mu >0$ is large and the last condition is replaced with $|\gamma'(s)| \leq m,$ where $m$ is some positive constant~\cite{2019}.
Also, in two dimensional case, the same result holds when $\gamma \in C^{3}((0, \infty))$ 
and $|\,\gamma'\,|+|\,\gamma''\,| \in L^{\infty}((0, \infty))$
~\cite{winkler2022}.
\\
Now,
 we write some important results related to the problem~\eqref{yazahraadrecni}.
The origin of the definition of this problem comes from 
the following chemotaxis-Navier–Stokes system which describes the motion of oxygen-driven swimming bacteria in an in-compressible
 fluid 
\begin{equation*}
	\begin{cases}
	u_{t}+ \omega \cdot \nabla u =\nabla\cdot\big( \nabla u-u \,\xi(v) \nabla v\big), & x\in\Omega, \ t>0, \\
	v_{t}+ \omega \cdot \nabla v=\Delta v-ug(v), & x\in\Omega, \ t>0,\\
	\omega_{t}+ (\omega \cdot \nabla) \omega =\Delta \omega -\nabla P+ u \nabla \phi, & x\in\Omega, \ t>0, \ t>0, \\
 \nabla \cdot \omega=0, & x\in\Omega, \ t>0, \ t>0. \\
	\end{cases}
	\end{equation*}
 Here, $u$ denotes the bacteria density and $v$ is the oxygen concentration. Also, $\omega$ and $P$ are the velocity and pressure of the
fluid, respectively. The function $ \xi$ measures the chemotactic sensitivity, $g$ is the consumption rate of the oxygen by the bacteria, and $\phi$
is a given potential function~\cite{2005}.
For the related results
with the chemotaxis-Navier–Stokes systems, we refer the interested readers to~\cite{2021w, 2014w,2016l, 2023} and references therein. We see that 
the problem~\eqref{yazahraadrecni} can be obtained from the preceding chemotaxis-Navier–Stokes system in the case of $\gamma(v)\equiv 1$ with the choice $\omega\equiv 0$ and $ g(v)=v.$
 For the problem~\eqref{yazahraadrecni},
in the absence of logistic source, when $\gamma(v) \equiv 1,\, \xi(v) \equiv \chi,$
where $\chi$ is some positive constant, for $n=2,$ the classical solutions are global and bounded in bounded convex domains with a smooth boundary~\cite{taowin}.
Also, for $n \geq 3,$ this result holds in bounded domains with a smooth boundary provided that
$
\|v_{0}\|_{L^{\infty}(\Omega)}\leq \frac{1}{6\,(n+1)\chi}$~\cite{tao}.
Later, this condition was extended to 
$\|v_{0}\|_{L^{\infty}(\Omega)}\leq \frac{\pi}{\chi\, \sqrt{2(n+1)}}$ 
and the same result was obtained in the absence of logistic source~\cite{baghaei-khelghati} and the presence of logistic source
~\cite{baghaei-khelghati-logistic}.\\
We now state the related results to the
problem~\eqref{yazahraadrecni}
in the absence of logistic source,
when the function $\gamma$ is not constant.
We begin by stating the results obtained in~\cite{GW} which is for the positive function $\gamma \in C^{0}([0, \infty))$
and
$ \xi=-\gamma'.$ For non-negative initial data from
 $(C^{0}(\overline{\Omega}))^{*} \times L^{\infty}(\Omega)$, this problem admits a global very weak solution in all dimensions, also, 
 for $\gamma \in C^{1}([0, \infty)),$ 
 the solutions stabilize toward a semi-trivial spatially homogeneous steady
state in the large time limit.
This result is obtained for $n\leq 3$ in~\cite{GW}
 and later for $n \geq 1$ in~\cite{ph}.
 If the decreasing function $\gamma $ belongs to $ C^{3}([0, \infty))$ and 
$ \xi=-\gamma',$ then for non-negative initial data from $C^{0}(\overline{\Omega}) \times W^{1,\infty}(\overline{\Omega})$ under a smallness assumption on $\|v_{0}\|_{L^{\infty}(\Omega)},$ there exists a unique global classical solution 
 that is bounded 
~\cite{6}.
If the initial data $(u_{0}, v_{0})$ belongs to $(W^{1,\infty}(\Omega))^{2}$ and $\gamma \in C^{3}([0, \infty)),$
 then classical solutions are globally bounded when $n\leq 2,$ and weak solutions are global
when $n \geq 3,$ in particular, such weak solutions become eventually smooth if $n = 3$~\cite{app}.
If $\gamma\in C^{0}([0, \infty))\cap C^{3}([0, \infty)),$
$\gamma(s)>0$
for all $s > 0$ as well as
$ \xi=-\gamma',$ and 
$\gamma$ satisfies:
\begin{equation*}
 \liminf_{s \searrow 0}
 \frac{\gamma(s)}{s^{\alpha}}>0
 \quad \mbox{and}
 \quad \limsup_{s \searrow 0} s^{\beta} \,|\gamma'(s)|< \infty
\end{equation*}
with $\alpha > 0$ and $\beta > 0,$
then this problem admits a global generalized solution for all reasonably regular initial data~\cite{w}. 
In~\cite{tw2023}, 
the function $\gamma$ is assumed to satisfy:
\begin{equation*}
 c_{1}\, s^{-k} \leq \gamma(s) \leq c_{2}\, s^{-k}
 \quad \mbox{for all} \quad s>0, \quad (*)
\end{equation*}
where $k,$ $c_{1}$ and $c_{2} $ are some positive constants. Under this assumption for
$\gamma \in C^{3}([0, \infty))$
and the initial data belongs to $(W^{1,\infty}(\Omega))^{2},$
the problem~\eqref{yazahraadrecni}
admits a global classical solution when $n=1,$ and a global weak-strong solution
when $n \geq 2.$ Also, this result is true if $2\leq n\leq 5,$ $k>\frac{n-2}{6-n}$ and
$\gamma$ satisfies $(*)$ and
\begin{equation*}
 |\gamma'(s)| \leq c_{3}\, s^{-k-1}
 \quad \mbox{for all} \quad s>0,
\end{equation*}
where $c_{3} $ is some positive constant~\cite{tw2023}.\\
Finally, we state the important results related to the
problem~\eqref{yazahraadrecni}
in the presence of logistic source.
If $\gamma \in C^{3}([0, \infty))$ and $\gamma'(s)<0$
for all $s \geq 0$ as well as
$ \xi=-\gamma',$ then
for $n = 2$ and $\mu >0,$ the solutions are global and bounded~\cite{5}. Also, the same result is obtained when $n \geq 3$ and $\mu$ is suitably large~\cite{5}. Moreover,
 the solution converges exponentially to $(1, 0)$ when $t$ tends to infinity~\cite{5}. For this problem,
 when the logistic source is as $f(u)=a\, u -\mu \,u^{\kappa}$ with $a>0, \, \mu>0$ and $\kappa>1,$ the classical solutions are global and bounded
 if one of the cases 
 $\big( n\leq 2, \kappa>1;$
 $ n\geq 3, \kappa>2$ or
 $ n\geq 3, \kappa=2$ and $\mu$ is large\big) holds~\cite{7}.
If the positive decreasing function $\gamma$ belongs to $ C^{2}([0, \infty))$ and
 $\gamma''\geq 0$ as well as $ \xi=-(1-\alpha)\,\gamma'$
 with $\alpha \in (0,1),$ in~\cite{4}, it is proved that the problem~\eqref{yazahraadrecni}
 admits a unique global classical solution that is uniformly
in time bounded provided that:
 \begin{align*}
 & \frac{(\gamma'(s))^{2}}{\gamma''(s)}
 \leq \frac{n}{2(n+1)^{3}}, \qquad 0< \|v_{0}\|_{L^{\infty}(\Omega)}\leq \gamma^{-1}\left(\frac{1}{n+1}\right)
 \end{align*} 
 and
 \begin{align*}
 \mu > \max_{0< s \leq \|v_{0}\|_{L^{\infty}(\Omega)}}
 \frac{-\gamma'(s)\, \|v_{0}\|_{L^{\infty}(\Omega)}}{\gamma(s)}.
 \end{align*}
 We see that the above conditions in the case of $\gamma(s)=(1+s)^{-k}\,(k>0)$ are as follows:
\begin{align*}
 k<\frac{n}{2(n+1)^{3}-n},
 \quad \quad
 \mu>k\|v_{0}\|_{L^{\infty}(\Omega)}
\qquad \mbox{and} \qquad 
 \|v_{0}\|_{L^{\infty}(\Omega)}\leq \gamma^{-1}\left(\frac{1}{n+1}\right). 
\end{align*}
Because of $\gamma'<0,$
the last condition is written as:
\begin{align*}
0< \|v_{0}\|_{L^{\infty}(\Omega)}\leq (n+1)^{\frac{1}{k}}-1.
\end{align*}
In~\cite{KHB}, we studied the special case
$\gamma(s)=(1+s)^{-k}\,(k>0)$ 
and we were able to improve the conditions in~\cite{4}. In fact, we proved the same result in~\cite{4} under the following conditions:
 \begin{align*}
k\,(1-\alpha)<\frac{4}{n+5}, \qquad 0<\|v_{0}\|_{L^{\infty}(\Omega)}\leq \Bigg[ \frac{4\big[l-1\big]}{ n+1}\Bigg]^{\frac{1}{k}}-1
 \end{align*}
and
\begin{equation*}
 \mu>
 \frac{n\|v_{0}\|_{L^{\infty}(\Omega)}}{l\,(n+1)(1+\|v_{0}\|_{L^{\infty}(\Omega)})}
 \end{equation*}
 with $l=\frac{1}{k\,(1-\alpha)}.$
In this paper, we focus again on the functions $\gamma$ and
$\xi$ as follows:
\begin{equation}\label{yarahim+}
 \gamma(s)=(1+s)^{-k}\quad
 \mbox{and} \quad
 \xi(s)=-(1-\alpha)\gamma'(s),
\end{equation}
where $k>0$ and $\alpha \in (0,1)$ 
 and extend our recent result. 
In fact, we remove the condition on $ k\,(1-\alpha)$
and prove the solutions are uniformly in time bounded under some conditions on $ \|v_{0}\|_{L^{\infty}(\Omega)}$ and $ \mu.$

\section{Our results}

Here, we state the standard well-posedness and classical solvability result.
\begin{lemma}\label{lem3}
 Let $u_{0} \geq 0$ and $v_{0}\geq 0$ satisfy $(u_{0}, v_{0} ) \in (W^{1,r}(\Omega))^{2}$ for some $r > n.$ Then problem~\eqref{yazahraadrecni} has a unique local in time
classical solution 
\begin{equation*}
 (u,v) \in \left(C \big([0, T_{\max}); W^{1,r}(\Omega)\big) \cap C^{2,1}\big(\,\overline{\Omega}\times (0, T_{\max})\right)^{2}
\end{equation*}
where $T_{\max} $ denotes the maximal existence time. In addition, if $T_{\max}< + \infty, $
then:
\begin{equation*}
 \limsup_{t\rightarrow T_{\max} }
 \|u(\ .\ ,t)\|_{L^{\infty}(\Omega)}= + \infty.
\end{equation*}
 Moreover, $u$ and $v$ satisfy the following inequalities:
\begin{equation}\label{yaallah1}
 u \geq 0 \quad \mbox{and} \quad 0 \leq v \leq \|v_{0}\|_{L^{\infty}(\Omega)}
 \quad \mbox{in} \quad \Omega\times (0, T_{\max}),
\end{equation}
also,
\begin{align}\label{mamnonam}
 \int_{\Omega} u(\ .\ , t) \, \dd x \leq c,
\end{align}
where $c$ is some positive constant.
\end{lemma}
For details of the proof, we refer the reader to~\cite{55, 4}.\\
Based on main ideas in~\cite{baghaei-khelghati,
baghaei-khelghati-logistic, KB1, KB2}, we write the following 
key lemma similar to~\cite[Lemma~2.2]{KHB}.
\begin{lemma}\label{lem1}
	Let $(u, v)$ be the solution of problem~\eqref{yazahraadrecni}. If there exists a smooth
positive function $\varphi(s)$ such that for
$0\leq s\leq \|v_{0}\|_{L^{\infty}(\Omega)},$
 the following inequality holds: 
\begin{equation}\label{yaallah}
(B(s))^{2}-4\, A(s)\, C(s) \leq 0,
\end{equation}
where for $p\geq 2,$ the functions $A, B$ and $C$ are defined as:
 \begin{equation}\label{313}
	\begin{cases}
	 A(s)=(p-1)\, \varphi (s)\, \gamma(s), \\
	B(s)=(p-1)\, \varphi (s)\, \xi(s) -\varphi' (s) \, (\gamma(s)+1), \\
 C(s)= \frac{1}{p}\, \varphi ''(s)- \varphi' ( s)\, \xi(s),
	\end{cases}
	\end{equation}
 then:
	\begin{equation*}
	 \frac{1}{p}\,\frac{\dd}{\dd t}\int _{\Omega} u^{p} \varphi (v) \, \dd x 
 \leq -\int _{\Omega} \left[\mu\, \varphi (v)+\frac{1}{p}\, v \, \varphi' (v)\right] u^{p+1}\, \dd x
 +\mu \int_{\Omega}u^{p} \varphi (v)\, \dd x.
	\end{equation*}
 \begin{proof}
 We assume that there exists
a smooth
positive function $\varphi(s)$ such that
for $0\leq s\leq \|v_{0}\|_{L^{\infty}(\Omega)}$ and $p\geq 2,$~\eqref{yaallah} holds. We take this function
and
 use~\eqref{yazahraadrecni} and integration by parts to write:
\begin{equation}
\label{yasahebazzaman} 
\begin{aligned}
 \frac{1}{p}\frac{\dd}{\dd t}\int _{\Omega} u^{p} \varphi (v) \, \dd x & =
 \int _{\Omega} u^{p-1} \varphi ( v)\, u_{t} \, \dd x
 + \frac{1}{p}\,\int_{\Omega}u^{p} \, \varphi' ( v) \, v_{t} \, \dd x %\nonumber
 \\
 &=
 - (p-1) \int_{\Omega} u^{p-2} \varphi ( v) \,\gamma(v) \, |\nabla u|^{2} \, \dd x
 %\nonumber
%\end{aligned} 
 \\
%\begin{aligned}
 & \hspace*{15mm}+ \int _{\Omega}u^{p-1} \left[\, (p-1)\, \varphi (v) \,\xi(v) -\varphi' (v) \, (\gamma(v)+1) \right]
 \big(\nabla u \cdot \nabla v\big) \, \dd x
 %\nonumber
 \\
 & \hspace*{15mm}+ \int _{\Omega}u^{p} \left[\, \varphi' ( v) \,\xi(v) - \frac{1}{p} \,\varphi ''(v ) \right] \,|\nabla v |^{2} \, \dd x 
 %\nonumber
 \\
 & \hspace*{15mm}- \int _{\Omega} \left[\,\mu\, \varphi (v)+\frac{1}{p} \,v \, \varphi' (v)\right] u^{p+1}\, \dd x
 +\mu \,\int_{\Omega}u^{p} \varphi ( v)\, \dd x.
 \end{aligned}
 \end{equation}
 For convenience in calculations, we write~\eqref{yasahebazzaman} as follows:
	\begin{equation}\label{yarahim}
	 \frac{1}{p}\,\frac{\dd}{\dd t}\int _{\Omega} u^{p} \varphi (v) \, \dd x 
	= \int _{\Omega} J (u, v) \, \dd x 
	- \int _{\Omega} \left[\,\mu\,\varphi (v)+\frac{1}{p} \, v \, \varphi' (v)\right] u^{p+1}\, \dd x
 +\mu \int_{\Omega} u^{p} \varphi ( v)\, \dd x
	\end{equation}
	with
\begin{equation}
	\begin{aligned}\label{ya mahdi adrecni}
	J(u, v)&=
 - (p-1)\, u^{p-2} \varphi (v)\, \gamma(v) \, |\nabla u|^{2} %\nonumber
 \\
 & \hspace*{15mm}
 +u^{p-1} \left[\, (p-1) \,\varphi (v) \,\xi(v) \,-\varphi' (v) \, (\gamma(v)+1) \right]
 \big(\nabla u \cdot \nabla v\big) 
 %\nonumber
 \\
 & \hspace*{15mm}
 + u^{p} \left[\, \varphi' ( v)\, \xi(v) - \frac{1}{p} \, \varphi ''(v ) \right] |\nabla v |^{2} %\nonumber
 \\
&=	 - u^{p-2} \, |\nabla u |^{2} A(v)
	+u^{p-1} 
	\big( \nabla u \cdot
	\nabla v \big) B(v)
	- u^{p} \,|\nabla v |^{2} \, C(v),
	\end{aligned}
	\end{equation} 
	where $ A, B $ and $ C $ are defined in~\eqref{313}.
	Now, by considering~\eqref{ya mahdi adrecni}, we can write
	\begin{align*}
J(u,v) 
& =-\bigg(\sqrt{ u^{p-2} A(v)}\, \nabla u
- \frac{u^{p-1}B(v)}{2\sqrt{ u^{p-2} A(v)}}\, \nabla v\bigg)\cdot\bigg(\sqrt{ u^{p-2} A(v)}\, \nabla u
-\frac{u^{p-1}B(v)}{2\sqrt{ u^{p-2} A(v)}}\, \nabla v\bigg)
\\
&\hspace*{78mm}+u^{p}\left[ \frac{(B(v))^{2}}{4 \, A(v)}-C(v)\right]|\nabla v |^{2}\quad \
 \\
&\leq 
u^{p}\left[ \frac{(B(v))^{2}-4\,A( v)\, C(v)}{4 \,A(v)} \right]|\nabla v |^{2}.
\end{align*}
	In view of the condition~\eqref{yaallah}, we see that 
$J\leq 0.$ Thus, the equality~\eqref{yarahim}
becomes
	\begin{equation*}
	 \frac{1}{p}\, \frac{\dd}{\dd t}\int _{\Omega} u^{p} \varphi (v) \, \dd x 
\leq -\int _{\Omega} \left[\,\mu\, \varphi (v)+\frac{1}{p}\, v \, \varphi' (v)\right] u^{p+1}\, \dd x +
 \mu \int_{\Omega}u^{p} \varphi ( v)\, \dd x.
	\end{equation*}
This completes our proof.
 \end{proof}
 \end{lemma}
 Before presenting a smooth positive function
 $\varphi$ such that the relation~\eqref{yaallah} holds,
 we present the following preliminary lemma.
\begin{lemma}\label{lemk}
Let the function $\gamma$ for $s\geq 0$ is defined as $\gamma(s)=(1+s)^{-k}$ with $k>0.$
If the function $H$ for $z>0$ is defined as
 \begin{equation}\label{1212}
 H(z) 
 = 
d^{2}\,p\,z^{2\lambda}-2\, d ^{2}(p-2)\, z^{2\lambda-1} +d^{2}p\,z^{2\lambda-2} %\\
 - 2\, d \big[2\, d \,(p-1)-p
 +2 \,l\big]z^{\lambda-1}
 -2\, d \, p\,z^{\lambda} +p,
\end{equation}
where for $\alpha \in (0,1)$ and $p>2,$ 
the parameters $\lambda, l$ and $d$ are as follows:
\begin{equation*}
 \lambda=d\,(p-1)\,(1-\alpha),\qquad l=\frac{1}{k\,(1-\alpha)} 
\end{equation*}
and
 \begin{align}\label{3132}
 d
 >\frac{-l+\sqrt{\,l^{2}+p\,(p-2)}}{2\,(p-2)}.
\end{align} 
Then there exists $\delta_{0}>0$
such that for 
$0<s\leq \delta_{0},$ $H(\gamma(s))\leq 0.$
\begin{proof}
 At first, we see that 
 \begin{align*}
 H(1) 
 &= 
d^{2}\,p-2\, d ^{2}(p-2) +d^{2}\,p
%\nonumber\\
% & \quad
 - 2\, d \big[2\, \dd \,(p-1)-p
 +2 \,l\big]
 -2\, d \,p+ p\\
 & =
- 4\,(p-2)\, d ^{2}-4 \,l\, d +p.
\end{align*}
It is not difficult to see that the choice of $d$ as 
\eqref{3132}
implies that:
\begin{align*}
- 4\,(p-2)\, d ^{2}-4 \,l\, d +p
 <0.
\end{align*}
Thus, $ H(1)<0.$
By considering
the continuity of the function $H$ on $(0, \infty)$ and $\gamma$ on $[0, \infty),$ and also
$ H(\gamma(0))=H(1)<0,$
we conclude that there exists $\delta_{0}>0$
such that for 
$0<s\leq \delta_{0},$ $H(\gamma(s))\leq 0.$
\end{proof}
\end{lemma}
 We now present a smooth positive function
 $\varphi$ and show that for this function, the relation~\eqref{yaallah} holds.
We must state that
the following lemma is the only place where the special choice of $\gamma$ is used.
\begin{lemma}\label{lem 2}
Assume that the initial values $u_{0}$
 and $ v_{0} $ are non-negative and
 satisfy: 
 \begin{equation*}
 (u_{0}, v_{0} ) \in (W^{1,r}(\Omega))^{2} 
 \quad \mbox{for some} \quad r > n,
 \end{equation*}
 and 
 \begin{equation*}
 0<\|v_{0}\|_{L^{\infty}(\Omega)}\leq \delta_{0},
 \end{equation*}
 which $\delta_{0}$ is introduced in Lemma~\ref{lemk}.
Also, assume that the functions $\gamma(v)$
and $\xi(v)$
	are defined as~\eqref{yarahim+}
 and
 for $p>n,$ the following condition
 holds:
 \begin{equation}
 \label{sallamkhoda}
 \mu>
 \frac{d\, (p-1)\, \|v_{0}\|_{L^{\infty}(\Omega)}}{l\,p\,(1+\|v_{0}\|_{L^{\infty}(\Omega)})},
 \end{equation}
 where $l=\frac{1}{k\,(1-\alpha)}$ 
 and $d$ is chosen as~\eqref{3132}.
Then, there exists some positive constant $c$ which also depends on $\delta_{0}$ and $p$ such that the following estimate holds:
\begin{equation}\label{313*}
\|u(\ .\ ,t)\|_{L^{p}(\Omega)}\leq c, \qquad \forall \,t \in (0, T_{\max}).
\end{equation}
 \begin{proof}
We prove this lemma in two steps. In the first step, we present a smooth positive function $\varphi$ and show that for this function~\eqref{yaallah} holds. 
In the second step, we prove~\eqref{313*} holds.


\begin{proof}[Step 1] 
 As a starting point, in Lemma~\ref{lem1}, we take
 $p > n$ and define the smooth positive function $\varphi$ as follows:
\begin{equation*}
 \varphi ( v)=\expe^{(\gamma(v))^{\lambda}}
\end{equation*} 
with $ \lambda=d\,(p-1)\,(1-\alpha),$
where $d$ is taken as~\eqref{3132}.
For this function, we have:
\begin{align}
	&\varphi'(v)
=\lambda \, (\gamma(v))^{\lambda-1}\gamma'(v) \,\varphi ( v)\label{127}
\end{align}
and
\begin{align*}
	&\varphi''(v)
=\lambda\,\big[\,(\lambda-1) (\gamma'(v))^{2} + \gamma(v)\gamma''(v) 
+\lambda \, (\gamma(v))^{\lambda} (\gamma'(v))^{2} \,\big](\gamma(v))^{\lambda-2}\varphi ( v).\nonumber
	\end{align*}
 Because of 
 \begin{align*}
 \gamma'(v)=-k\,(1+v)^{-k-1}
 \quad \mbox{and}
 \quad 
 \gamma''(v)=k\,(k+1)\,(1+v)^{-k-2},
 \end{align*}
 we see that $ \gamma(v) \gamma''(v)=\big(1+\frac{1}{k}\big)(\gamma'(v))^{2}.$
 Thus,
 \begin{equation}
	\varphi''(v)
=\lambda\left[\lambda+\frac{1}{k}
+\lambda \,(\gamma(v))^{\lambda} \right](\gamma(v))^{\lambda-2}\,(\gamma'(v))^{2} \varphi ( v).\label{125}
	\end{equation}
 By considering the values of $A, B$ and $C$ from~\eqref{313}, we can write:
\begin{multline*}
 (B(v))^{2}-4 \,A( v) \,C(v) %\nonumber\\
 =(p-1)^{2} (\varphi(v)) ^{2} \, (\xi (v)) ^{2}+
 (\varphi' (v)) ^{2} \, (\gamma(v)+1)^{2} \nonumber
 \\
 \quad
 -\frac{4 \,(p-1)}{p}\, \varphi ( v) \, \varphi ''(v )\, \gamma(v)
 -2\,(p-1)\,\varphi ( v)\, \varphi' ( v)\, \xi(v) \, (1-\gamma(v)).
\end{multline*}
Making use of~\eqref{127},~\eqref{125} and $\xi(v)=-(1-\alpha)\gamma'(v),$
we obtain:
\begin{multline*}
 (B(v))^{2}-4\, A( v)\, C(v)%\\
 =(\gamma'(v))^{2}(\varphi(v)) ^{2}\left\{(p-1)^{2}(1-\alpha)^{2} +
 \lambda^{2} (\gamma(v))^{2\lambda-2} (\gamma(v)+1)^{2}
 \vphantom{\frac{4\,\lambda\, (p-1)}{p}}\right.\\
 %\quad
 \left. -\frac{4\,\lambda\, (p-1)}{p} \,\left[\lambda+\frac{1}{k}
+\lambda (\gamma(v))^{\lambda} \right](\gamma(v))^{\lambda-1}\, 
%\\& \quad
 +2\,\lambda\,(p-1)(1-\alpha) (\gamma(v))^{\lambda-1} (1-\gamma(v))\right\}.
 \end{multline*}
 Because of $ \lambda=d\,(p-1)(1-\alpha),$
 we replace $(p-1)(1-\alpha)$ with $\frac{\lambda}{d}$
 to have:
 \begin{multline}\label{11012}
 (B(v))^{2}-4\, A( v)\, C(v)\\
 \begin{aligned}
 {}={}&(\gamma'(v))^{2}(\varphi(v)) ^{2}\Bigg\{\frac{\lambda^{2}}{d^{2}} +
 \lambda^{2} (\gamma(v))^{2\lambda-2} (\gamma(v)+1)^{2}\\
 & \quad
 -\frac{4\,\lambda^{2}}{d\,p\,(1-\alpha)}
 \,\left[\lambda \, \big(1+(\gamma(v))^{\lambda}\big)+\frac{1}{k}
 \right](\gamma(v))^{\lambda-1}
 +\frac{ 2\,\lambda^{2} }{d} (\gamma(v))^{\lambda-1} (1-\gamma(v))\Bigg\}
\\
 {}={}&\frac{\lambda^{2}(\gamma'(v))^{2}(\varphi(v)) ^{2} }{d^{2}\, p}\Bigg\{ p + d^{2}\, p\,
 (\gamma(v))^{2\lambda-2} (\gamma(v)+1)^{2}\\
 & \quad 
 -4\, \dd \,\left[d\,(p-1)(1+(\gamma(v))^{\lambda})+\frac{1}{k\,(1-\alpha)} \right](\gamma(v))^{\lambda-1}
 +2\, \dd \,p\, (\gamma(v))^{\lambda-1} (1-\gamma(v))\Bigg\}\, \\
 {}={}&\frac{\lambda^{2} (\gamma'(v))^{2}(\varphi(v)) ^{2}}{d^{2}\, p}\Bigg\{p +
\, \dd ^{2}p\,(\gamma(v))^{2\lambda}-2\, \dd ^{2}(p-2) (\gamma(v))^{2\lambda-1} \\
 & \quad
 +d^{2}\,p\,(\gamma(v))^{2\lambda-2} - 2\, \dd \big[2\, \dd \,(p-1)-p
 +2 \,l\big](\gamma(v))^{\lambda-1}
 -2\, \dd \,p\,(\gamma(v))^{\lambda}\Bigg\}\\
 {}\coloneqq{} &\frac{\lambda^{2}}{d^{2}\, p}(\gamma'(v))^{2} (\varphi(v)) ^{2}
 H(\gamma (v)),
\end{aligned}
\end{multline}
where the function $H$ is defined by~\eqref{1212}.
By considering Lemma~\ref{lemk},
we see that there exists $\delta_{0}>0$
such that for 
$0<v\leq \delta_{0},$ $H(\gamma(v))\leq 0.$ Because of
$0<\|v_{0}\|_{L^{\infty}(\Omega)}\leq \delta_{0},$ we conclude that $H(\gamma(v))$ is non-positive in the interval $(0,\|v_{0}\|_{L^{\infty}(\Omega)}]. $ 
 Therefore, by
combining this with~\eqref{11012}, 
 we conclude that
 the relation~\eqref{yaallah} holds. 
\let\qed\relax
\end{proof}
\begin{proof}[Step 2]
 In order to prove~\eqref{313*}, we apply Lemma~\ref{lem1} and write:
\begin{multline}\label{sallamfrmandeh}
	 \frac{1}{p}\,\frac{\dd}{\dd t}\int _{\Omega} u^{p} \varphi (v) \, \dd x +\mu \int_{\Omega}u^{p} \varphi ( v)\, \dd x %\nonumber
	 \\
 \leq -\int _{\Omega} \left[\mu\, \varphi (v)+\frac{1}{p}\, v \, \varphi' (v)\right] u^{p+1}\, \dd x +2\,\mu \int_{\Omega}u^{p} \varphi ( v)\, \dd x.
	\end{multline}
We now apply the Young inequality to the second term on the right hand side of~\eqref{sallamfrmandeh} to have: 
 \begin{align*}
 2\,\mu \int_{\Omega}u^{p} \, \varphi ( v)\, \dd x 
 &
 \leq \epsilon \int_{\Omega}u^{p+1} \varphi ( v)\, \dd x\, 
 + c_{\epsilon} \int_{\Omega}\varphi ( v)\, \dd x, 
 \end{align*}
 where $\epsilon$ is chosen as follows:
 \begin{equation}\label{yamahdi}
 0< \epsilon <\mu-
 \frac{ k\,\lambda \,\|v_{0}\|_{L^{\infty}(\Omega)}}{p\, (1+\|v_{0}\|_{L^{\infty}(\Omega)})}
 \end{equation}
 and:
 \begin{equation*}
 c_{\epsilon}=\frac{1}{p+1}\,\left[\frac{p}{\epsilon\,(p+1)}\right]^{p}(2\mu)^{p+1}.
 \end{equation*}
 Because of $\gamma'(v)<0,$~\eqref{127} implies that $\varphi' ( v)<0.$ Thus, 
 \begin{align}\label{yaali}
 \varphi (\|v_{0}\|_{L^{\infty}(\Omega)}) < \varphi ( v)<\varphi (0).
 \end{align}
 Making use of this, we can write:
 \begin{align}\label{shokr}
 2\,\mu \int_{\Omega}u^{p} \varphi ( v)\, \dd x 
 &
 \leq \epsilon \int_{\Omega}u^{p+1} \varphi ( v)\, \dd x+ c_{0} 
 \end{align}
 with $c_{0} =c_{\epsilon}\,|\Omega| \,\varphi (0).$
 We now combine the inequality~\eqref{shokr}
 with~\eqref{sallamfrmandeh}
 to obtain:
 \begin{equation}\label{khodayakomakamkon}
	\frac{1}{p}\, \frac{\dd}{\dd t}\int _{\Omega} u^{p} \varphi (v) \, \dd x +\mu \int_{\Omega} u^{p} \varphi ( v)\, \dd x %\nonumber
%	\\
 \leq \int _{\Omega} \left[(\epsilon-\mu) \,\varphi (v)-\frac{1}{p}\, v \, \varphi' (v)\right] u^{p+1}\, \dd x
+ c_{0}.
	\end{equation}
We use $\gamma'(v)=-\frac{k}{1+v} \gamma(v), \, \gamma(v)\leq 1$ and~\eqref{yamahdi} to have
 \begin{align*}
 (\epsilon-\mu)\, \varphi (v)-\frac{1}{p} v \, \varphi' (v)
 &=\left[
 \epsilon-\mu-\frac{ \lambda }{p}\, v \, (\gamma(v))^{\lambda-1}\gamma'(v)\right] \varphi ( v)\\
 &=
 \left[
 \epsilon-\mu+\frac{ k\,\lambda\, v }{p\,(1+v)}\, \, (\gamma(v))^{\lambda}\right] \varphi ( v) \\
 & \leq
 \left[ \epsilon-\mu+\frac{ k\,\lambda \,\|v_{0}\|_{L^{\infty}(\Omega)}}{p\, (1+\|v_{0}\|_{L^{\infty}(\Omega)})}\, \right] \varphi ( v)\\
 & \leq 0.
 \end{align*} 
 This along with~\eqref{khodayakomakamkon} yields:
 \begin{equation*}
	 \frac{\dd}{\dd t}\int _{\Omega} u^{p} \varphi (v) \, \dd x +\mu\,p\int_{\Omega}u^{p} \varphi ( v)\, \dd x
 \leq
p\,c_{0}.
	\end{equation*}
 We put: 
 \begin{equation*}
 y(t)=\int _{\Omega} u^{p} \varphi (v) \, \dd x.
 \end{equation*}
 Thus,
 \begin{align*}
y'(t)
 +\mu\,p \,y(t)
 \leq
 p\,c_{0}.
	\end{align*}
 This yields:
 \begin{align}\label{14}
y(t)
 \leq \max\Big\{y(0), \frac{c_{0}}{\mu}
\Big\}.
	\end{align}
This along with~\eqref{yaali} allows us to write:
 \begin{equation*}
 \int _{\Omega} u^{p} \, \dd x \leq
 \big( \varphi (\|v_{0}\|_{L^{\infty}(\Omega)})\big)^{-1} \max\Big\{y(0), \frac{c_{0}}{\mu }
\Big\}\coloneqq c.
 \end{equation*}
By considering the value of $c_{0}, $ we see that $c$ also depends on $\delta_{0}$ and $p.$
 This completes our proof.
\end{proof}
\let\qed\relax
\end{proof}
\end{lemma}
 The proof of the following lemma is the same as~\cite[Lemma~3.2]{tao} or~\cite[Lemma~2.4]{KHB}. 
 But, we write it to complement our content.
 \begin{lemma}\label{lem5}
Assume that for $p>n,$ the following estimate
 holds:
\begin{equation}\label{77}
\|u(\ .\ ,t)\|_{L^{p}(\Omega)}\leq C_{1}, \qquad \forall \,t \in (0, T_{\max}),
\end{equation}
where $C_{1}$ is some positive constant, then there exists some positive constant $C_{2}$ which also depends on $C_{1}$ 
 and $
 \|v_{0}\|_{L^{\infty}(\Omega)}$ 
 such that
\begin{equation}\label{313*****}
 \|\nabla v(\ .\ ,t)\|_{L^{\infty}(\Omega)} \leq C_{2}
\end{equation}
for all $t \in (0, T_{\max}).$
\begin{proof}
By considering 
Lemma~\ref{lem3}, we see that it is sufficient to prove 
for any $\tau \in (0, T_{\max}),$ 
\begin{equation}\label{110}
 \|\nabla v(\ .\ ,t)\|_{L^{\infty}(\Omega)} \leq C_{2} \quad \mbox{for all} \ t \in (\tau, T_{\max}).
\end{equation}
We use the representation formula for the second equation~\eqref{yazahraadrecni}
to have:
\begin{equation*}
v(\ .\ ,t)=\textrm{e}^{t(\Delta-1)}v_{0}+\int_{0}^{t}\textrm{e}^{(t-s)(\Delta-1)}(1-u(\ .\ ,s))\, v(\ .\ ,s) \, \dd s, \quad t \in (0, T_{\max}).
\end{equation*}
We now take $p>n$ and use 
$0\leq v\leq
 \|v_{0}\|_{L^{\infty}(\Omega)}$ to write:
\begin{equation}\label{5}
\begin{aligned}
 \|(1-u(\ .\ ,s))\, v(\ .\ ,s)\|_{L^{p}(\Omega)}
 &\leq
 \|v(\ .\ ,s)\|_{L^{\infty}(\Omega)} \left(\int_{\Omega} |1-u(\ .\ ,s))|^{p} \, \dd x\right)^{\frac{1}{p}} %\nonumber
 \\
 &
 \leq
 \|v(\ .\ ,s)\|_{L^{\infty}(\Omega)} \left(\int_{\Omega} \big(1+|u(\ .\ ,s))|\,\big)^{p} \, \dd x\right)^{\frac{1}{p}} %\nonumber
 \\
 & \leq
 \|v(\ .\ ,s)\|_{L^{\infty}(\Omega)} \left(2^{p-1}
 \int_{\Omega} \big(1+|u(\ .\ ,s))|^{p}\,\big) \, \dd x\right)^{\frac{1}{p}}
% \nonumber
\\
 & \leq
 2^{\frac{p-1}{p}} \|v(\ .\ ,s)\|_{L^{\infty}(\Omega)} \left(|\Omega|^{\frac{1}{p}}
 +\|u(\ .\ ,s))\|_{L^{p}(\Omega)}\,\right) 
% \nonumber
\\
 & \leq c,
\end{aligned}
\end{equation}
where we have used the inequality $(a + b)^{m} \leq 2^{m-1}(a^{m}+b^{m})$ with $a, b \geq 0$ and $m > 1,$ also $(a + b)^{m'} \leq (a^{m'}+b^{m'})$
with $0<m'<1.$
We note that the constant $c$ in the last estimate,
it also
depends on $C_{1}$ and $
 \|v_{0}\|_{L^{\infty}(\Omega)}.$ 
In order to prove~\eqref{110},
we take $ \tau\in (0, \min \{1,T_{\max}\})$ and $\theta\in (\frac{p+n}{2\,p}, 1)$ 
and use the estimates $(3.16)$ and $(3.17)$ in~\cite{tao},
also~\eqref{5} to obtain:
\begin{align*}\label{328}
\|v(\ .\ ,t)\|_{W^{1,\infty}(\Omega)}
&\leq
c\, \|(-\Delta+1)^{\theta}v(\ .\ ,t)\|_{L^{p}(\Omega)}\nonumber\\
&\leq
c \,t^{-\theta}\textrm{e}^{-\delta t}\,\|v_{0}\|_{L^{p}(\Omega)}+
c\int_{0}^{t}(t-s)^{-\theta}\textrm{e}^{-\delta\, (t-s)}\|(1-u(\ .\ ,s))\, v(\ .\ ,s)\|_{L^{p}(\Omega)}\, \dd s\nonumber\\
&\leq 
c \, t^{-\theta}+c\int_{0}^{t}(t-s)^{-\theta}\textrm{e}^{-\delta\,(t-s)} \, \dd s\nonumber\\
&\leq
c\, t^{-\theta}+c \int_{0}^{+\infty}\sigma^{-\theta}\textrm{e}^{-\delta\, \sigma} \, \dd \sigma \nonumber\\
&\leq 
c\, (\tau^{-\theta}+1), \quad t \in (\tau, T_{\max}),
\end{align*}
where the constants can vary from line to line.
In this estimate, from the third line on-wards, the constants are also dependent on $C_{1}$
 and $
 \|v_{0}\|_{L^{\infty}(\Omega)}.$ 
This completes our proof.
\end{proof}
\end{lemma}
The proof of the following lemma is the same as~\cite[Lemma~2.5]{KHB} and similar to~\cite[Lemma~3.2]{tao}.
 But, we write it to complement our content.
 \begin{lemma}\label{lem4}
Assume that the initial values $u_{0}$
 and $ v_{0} $ are non-negative and
 satisfy: 
 \begin{equation*}
 (u_{0}, v_{0} ) \in (W^{1,r}(\Omega))^{2} 
 \quad \mbox{for some} \quad r > n.
 \end{equation*}
Also, assume that~\eqref{313*****} holds.
Then there exists some positive constant $c$ 
 such that 
for all $t \in (0, T_{\max}),$ the following
estimate holds
\begin{equation*}
\|u(\ .\ ,t)\|_{L^{\infty}(\Omega)}\leq c.
\end{equation*}
\begin{proof}
 We take $q\geq 2$ and use from~\eqref{yazahraadrecni} and integration by parts to obtain:
\begin{equation}\label{313***} 
\begin{aligned}
\frac{\dd}{\dd t}\int_{\Omega} u^{q} \, \dd x
&= q\int_{\Omega} u^{q-1} \left[\nabla\cdot( \gamma(v)\, \nabla u-u\, \xi(v)\,\nabla v)+ \mu\, u \,(1- u) \right]\, \dd x 
%\nonumber
\\
&
=-q\,(q-1)\int_{\Omega}\gamma(v)\, u^{q-2} | \nabla u|^{2} \, \dd x + q\,(q-1) \int_{\Omega} u^{q-1} \xi(v)\, \big(\nabla u\cdot \nabla v\big) \, \dd x%\nonumber
\\
& \hspace*{75mm}%\quad
+ \mu\, q \,\int_{\Omega} u^{q} \,(1- u) \, \dd x.
\end{aligned}
\end{equation}
Because of $0\leq v\leq
 \|v_{0}\|_{L^{\infty}(\Omega)}, $
we have:
\begin{align*}
 & \gamma(v)=(1+v)^{-k}\geq (1+ \|v_{0}\|_{L^{\infty}(\Omega)})^{-k}\coloneqq c_{1},\\
 & \xi(v)=k\,(1-\alpha)(1+v)^{-k-1}\leq k(1-\alpha)\coloneqq c_{2}.
\end{align*}
Making use of these,~\eqref{313*****} and Young's inequality, we can write~\eqref{313***} as follows:
\begin{equation}\label{ya mahdi}
\begin{aligned}
\frac{\dd}{\dd t}\int_{\Omega} u^{q} \, \dd x 
&\leq- c_{1}\,q\,(q-1)\int_{\Omega} u^{q-2} | \nabla u|^{2} \, \dd x + C_{2} \,c_{2} \,q\,(q-1)\int_{\Omega} u^{q-1} |\nabla u| \, \dd x 
%\nonumber
%\\
%&\hspace*{70mm} %\qquad
+\mu\, q\,\int_{\Omega} u^{q}\, \dd x
%\nonumber
\\
&%\quad 
= 
-\frac{4 \,c_{1}\,(q-1)}{q}\int_{\Omega}\big|\nabla u^{\frac{q}{2}}\big|^{2} \, \dd x +2\, C_{2} \,c_{2} \,(q-1)\int_{\Omega} u^{\frac{q}{2}}\cdot \big|\nabla u^{\frac{q}{2}}\big| \, \dd x 
%\nonumber
%\\
%&\hspace*{70mm}
+\mu\, q\,\int_{\Omega} u^{q}\, \dd x
%\nonumber
\\
&%\quad 
\leq-\frac{2\,c_{1}\,(q-1)}{q}\int_{\Omega}\big|\nabla u^{\frac{q}{2}}\big|^{2} \, \dd x+q\,\left(\frac{(q-1)\,(C_{2} \, c_{2}) ^{2}}{2\,c_{1}}+\mu \right) \int_{\Omega} u^{q}\, \dd x.
\end{aligned}
\end{equation}
We now add $q \int_{\Omega} u^{q}\, \dd x$ on both sides of~\eqref{ya mahdi} to have:
\begin{equation}\label{main essential}
\frac{\dd}{\dd t}\int_{\Omega} u^{q} \, \dd x + q \int_{\Omega} u^{q}\, \dd x \leq 
-\frac{2\,c_{1}\,(q-1)}{q}\int_{\Omega}|\nabla u^{\frac{q}{2}}|^{2} \, \dd x + c_{3}\int_{\Omega} u^{q}\, \dd x
\end{equation}
 with 
 \begin{equation*}
 c_{3}=q\,\left(\frac{(q-1)\,(C_{2} \, c_{2}) ^{2}}{2\,c_{1}}+\mu +1\right).
 \end{equation*}
We first apply the Gagliardo--Nirenberg inequality and then
use the Young inequality with exponents $s=\frac{n+2}{2}$ and $s'=\frac{n+2}{n}$
 to obtain:
\begin{align*}
c_{3}\int_{\Omega} u^{q}\, \dd x
=c_{3} \, \big\|u^{\frac{q}{2}}\big\|_{L^{2}(\Omega)}^{2}
&\leq 
c_{3} \,(C_{GN})^{2}\left(\big\|\nabla u^{\frac{q}{2}}\big\|_{L^{2}(\Omega)}^{\frac{n}{n+2}}\big\| u^{\frac{q}{2}}\big\|_{L^{1}(\Omega)}^{\frac{2}{n+2}}+\big\| u^{\frac{q}{2}}\big\|_{L^{1}(\Omega)}\right)^{2}\\
&\leq 
2\, c_{3} \,(C_{GN})^{2}\left(\big\|\nabla u^{\frac{q}{2}}\big\|_{L^{2}(\Omega)}^{\frac{2n}{n+2}}\big\| u^{\frac{q}{2}}\big\|_{L^{1}(\Omega)}^{\frac{4}{n+2}}+\big\| u^{\frac{q}{2}}\big\|_{L^{1}(\Omega)}^{2}\right)\\
&\leq
\frac{2\,c_{1}\,(q-1)}{q}\big\|\nabla u^{\frac{q}{2}}\big\|_{L^{2}(\Omega)}^{2}+\big(c_{4}+2\, c_{3}\, (C_{GN})^{2}\big)\big\| u^{\frac{q}{2}}\big\|_{L^{1}(\Omega)}^{2}\\
&
=\frac{2\,c_{1}(q-1)}{q}\int_{\Omega}|\nabla u^{\frac{q}{2}}|^{2}\, \dd x +c_{5}\left(\int_{\Omega} u^{\frac{q}{2}}\, \dd x\right)^{2}
\end{align*}
with
\begin{align*}
 c_{4}=\frac{1}{s}\left(\frac{2\,c_{1}\, s'\, (q-1)}{q}\right)^{-\frac{s}{s'}}\big(2\, c_{3} \,(C_{GN})^{2}\big)^{s}
 \quad \mbox{and}\quad c_{5}=c_{4}+2\, c_{3}\, (C_{GN})^{2},
 \end{align*}
 where $C_{GN}$ is the constant in the Gagliardo–Nirenberg inequality.
 Combining the last inequality with~\eqref{main essential}
 yields:
\begin{equation*}
\frac{\dd}{\dd t}\int_{\Omega} u^{q} \, \dd x +q \int_{\Omega} u^{q}\, \dd x \leq 
 c_{5} \left(\int_{\Omega} u^{\frac{q}{2}}\, \dd x\right)^{2}.
\end{equation*}
For $0 \leq t \leq T_{\max}, $ we can write:
\begin{equation*}
\frac{\dd}{\dd t}\left(\expe^{q\,t}\int_{\Omega} u^{q} \, \dd x \right) \leq c_{5}\, \expe^{q\,t}\,
 \left(\int_{\Omega} u^{\frac{q}{2}}\, \dd x\right)^{2}.
\end{equation*}
Now, we integrate and use 
$\expe^{-q\,t}\leq 1$
to get:
\begin{align*}
\int_{\Omega} u^{q}\, \dd x &
\leq \int_{\Omega} u^{q}_{0} \, \dd x+
\frac{c_{5}}{q} \sup_{0\leq t\leq T_{\max}}\left(\int_{\Omega} u^{\frac{q}{2}}\, \dd x\right)^{2}\nonumber\\
 &
\leq |\Omega|\, \| u_{0}\|_{L^{\infty}(\Omega)}^{q}+
\frac{c_{5}}{q} \sup_{0\leq t\leq T_{\max}}\left(\int_{\Omega} u^{\frac{q}{2}}\, \dd x\right)^{2}.
\end{align*}
Thus,
\begin{align}\label{main essential11}
\left(\int_{\Omega} u^{q} \, \dd x\right)^{\frac{1}{q}} 
 &
\leq \left[\,|\Omega|\, \| u_{0}\|_{L^{\infty}(\Omega)}^{q}+
 \frac{c_{5}}{q} \sup_{0\leq t\leq T_{\max}}\left(\int_{\Omega} u^{\frac{q}{2}}\, \dd x\right)^{2}\right]^{\frac{1}{q}}\nonumber\\
 &
\leq |\Omega|^{\frac{1}{q}}\, \| u_{0}\|_{L^{\infty}(\Omega)}+
\big(\frac{c_{5}}{q}\big)^{\frac{1}{q}} \sup_{0\leq t\leq T_{\max}}\left(\int_{\Omega} u^{\frac{q}{2}}\, \dd x\right)^{\frac{2}{q}}.
\end{align}
We note that
\begin{align*}
 c_{5}&=c_{4}+2\, c_{3} \,(C_{GN})^{2} \\
 &=\frac{1}{s}\left(\frac{2\,c_{1}\, s'\, (q-1)}{q}\right)^{-\frac{s}{s'}}\big(2\, c_{3} \,(C_{GN})^{2}\big)^{s}+2\, c_{3} \, (C_{GN})^{2}\\
 &=\frac{1}{s}\big(2\,c_{1}\, s'\big)^{-\frac{s}{s'}}\big(2
 \,(C_{GN})^{2}\big)^{s}
 \big(\frac{q}{q-1}\big)^{\frac{n}{2}}(c_{3})^{s}+2\, c_{3}\, (C_{G N})^{2}\\
 & \leq m \big[\big(\frac{q}{q-1}\big)^{\frac{n}{2}}(c_{3})^{s}+c_{3} \big]\\
 & \leq m \big[\big(\frac{q}{q-1}\big)^{\frac{n}{2}}+1 \big](c_{3})^{s}
\end{align*}
with
 \begin{equation*}
 m=\max\Big\{\frac{1}{s}\,\big(2\,c_{1}\, s' \,\big)^{-\frac{s}{s'}}\big(2
 \,(C_{GN})^{2}\big)^{s},2\, (C_{GN})^{2}\Big\}.
 \end{equation*}
 Here, we have used from $c_{3}>1$ and $s>1.$
 By inserting $c_{3}$ and using $q \geq 2,$
 we obtain:
\begin{equation}\label{salam}
\begin{aligned}
 \frac{c_{5}}{q}&\leq m \left[\big(\frac{q}{q-1}\big)^{\frac{n}{2}}+1 \right]\left(\frac{(q-1)\,(C_{2}\,c_{2}) ^{2}}{2\,c_{1}}+\mu +1\right)^{\frac{n}{2}+1} q^{\frac{n}{2}}%\nonumber
 \\
 & \leq 
 2\, m \,\left(\frac{ (C_{2}\,c_{2})^{2}}{2\,c_{1}}+\mu +1\right)^{\frac{n}{2}+1}\, \big(\frac{q}{q-1}\big)^{\frac{n}{2}} (q-1)^{\frac{n}{2}+1}\, q^{\frac{n}{2}}%\nonumber
 \\
 &= c_{6} \, (q-1)\, q^{n} %\nonumber
 \\
 & \leq c_{6} \, q^{n+1}
\end{aligned}
\end{equation}
with
\[
 c_{6}= 2\, m \,\left(\frac{ (C_{2}\,c_{2}) ^{2}}{2\,c_{1}}+\mu +1\right)^{\frac{n}{2}+1}. 
\]
Making use of~\eqref{salam} and $q^{\frac{n+1}{q}}>1,$ we can write 
\eqref{main essential11} as follows:
\begin{equation}\label{yalatif}
\begin{aligned}
\left(\int_{\Omega} u^{q} \, \dd x\right)^{\frac{1}{q}} 
 &
\leq |\Omega|^{\frac{1}{q}}\, \| u_{0}\|_{L^{\infty}(\Omega)}+
\big( c_{6} \, q^{n+1}\big)^{\frac{1}{q}} \sup_{0\leq t\leq T_{\max}}\left(\int_{\Omega} u^{\frac{q}{2}}\, \dd x\right)^{\frac{2}{q}}
%\nonumber
\\
&
\leq c_{7}^{\frac{1}{q}} \, q^{\frac{n+1}{q}} \left( \| u_{0}\|_{L^{\infty}(\Omega)}+
 \sup_{0\leq t\leq T_{\max}}\left(\int_{\Omega} u^{\frac{q}{2}}\, \dd x\right)^{\frac{2}{q}}\right)
\end{aligned}
\end{equation}
with $c_{7}=|\Omega|+ c_{6}.$
We now define:
\begin{equation*}
M(q)=\max\left\lbrace \| u_{0}\|_{L^{\infty}(\Omega)}, \sup_{0\leq t\leq T_{\max}}\left(\int_{\Omega} u^{q}\, \dd x\right)^{\frac{1}{q}}\right\rbrace. 
\end{equation*}
This allows us to write~\eqref{yalatif} as:
\begin{equation*}
M(q)\leq 2\, c_{7}^{\frac{1}{q}} \, q^{\frac{n+1}{q}} M\left(\frac{q}{2}\right).
\end{equation*}
We now take $q= 2^{i}$
$(i\in\mathbb{N})$ to obtain:
\begin{equation}\label{yakarim}
\begin{aligned}
M(2^{i}) & \leq 2\, c_{7}^{2^{-i}}
 \, 2^{\frac{(n+1) i}{2^{i}}} M(2^{i-1}) %\nonumber
 \\
& \leq 2\, c_{7}^{2^{-i}+2^{-i+1}}
 2^{(n+1)\big(\frac{i}{2^{i}}+\frac{i-1}{2^{i-1}}\big)} 
M(2^{i-2}) %\nonumber
\\
& \leq \cdots %\nonumber 
\\
& \leq 2\, c_{7}^{2^{-i}+2^{-i+1}+\cdot+2^{-1}}
 2^{(n+1)\big(\frac{i}{2^{i}}+\frac{i-1}{2^{i-1} }+\cdot+\frac{1}{2}\big)} 
M(1),
\end{aligned}
\end{equation} 
and compute the following elementary series:
\begin{align*}
 S\coloneqq \sum_{i=1}^{\infty}\frac{i}{2^{i}}&= \sum_{i=0}^{\infty}\frac{i+1}{2^{i+1}}=\sum_{i=0}^{\infty}\big(\frac{i}{2^{i+1}}+\frac{1}{2^{i+1}}\big)=\frac{1}{2}\sum_{i=1}^{\infty}\frac{i}{2^{i}}+\sum_{i=0}^{\infty}\frac{1}{2^{i+1}}=\frac{1}{2}S+1.
\end{align*}
Thus, $S=2.$ Making use of this, $\lim_{i \rightarrow \infty}\| u(\ .\ ,t)\|_{L^{2^{i}}(\Omega)}=\|u(\ .\ ,t)\|_{L^{\infty}(\Omega)} $
and~\eqref{mamnonam}, by letting $i \rightarrow \infty$ in~\eqref{yakarim},
we obtain the desired result.
\end{proof}
\end{lemma}
We now can write our main theorem.
\begin{theo}	\label{thm}
 Let the initial values $u_{0}$
 and $ v_{0} $ are non-negative and
 satisfy: 
 \begin{equation*}
 (u_{0}, v_{0} ) \in (W^{1,r}(\Omega))^{2} 
 \quad \mbox{for some} \quad r > n
 \end{equation*}
 and 
 \begin{equation*}
 0<\|v_{0}\|_{L^{\infty}(\Omega)}\leq \delta_{0},
 \end{equation*}
 which $\delta_{0}$ is introduced in Lemma~\ref{lemk}.
 Assume that the condition~\eqref{sallamkhoda} holds. 
Then, 
 the solution of the problem~\eqref{yazahraadrecni} with the functions $\gamma(v)$
and $\xi(v)$
	 defined by~\eqref{yarahim+}
 is global and bounded.
\begin{proof}
 By considering the extensibility criterion provided by Lemma~\ref{lem3}, the proof is a consequence of~\eqref{yaallah} and Lemma~\ref{lem4}.\\
 \end{proof}
\end{theo}


\section*{Declaration of interests}
The authors do not work for, advise, own shares in, or receive funds from
any organization that could benefit from this article, and have declared no
affiliations other than their research organizations.

\printbibliography


\end{document}
 \small

\end{document}

